Author’s response to reviews

Title: Assessing the training needs of medical students in patient information gathering; a qualitative study

Authors:
Conor Gilligan (conor.gilligan@newcastle.edu.au)
Sonja Brubacher (s.brubacher@griffith.edu.au)
Martine Powell (martine.powell@griffith.edu.au)

Version: 1 Date: 22 Dec 2019

Author’s response to reviews:

Dear Editor,

Thank you for the opportunity to revise this paper. The reviewers made very valuable comments which we have addressed individually in the table at the beginning of the revised manuscript, and throughout the paper. These changes have improved the paper and we are grateful for the opportunity.

Kind regards,

Conor Gilligan on behalf of the authors.